 Vc/17-Feb-05
Week 07 starting 14 Feb 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 14 feb 05 @  2/14/2005 4:00:09 PM    PM    

	LHC SM18 Magnet Test Facility 


	6
	A1
	MBAL2123
	PT 13.1 IAP @ Warm
	294.46

	7
	A2
	MBAR3278
	Prep 5 Pumping + Cool Down
	157.95

	8
	B1
	SSS034
	PT 12 Warm Up
	300.28

	3
	B2
	MBAL3165
	PT 6.1 Training 1
	4.33 HV

	9
	C1
	MBAR3286
	Finished Ready to go Final scan
	295.8

	11
	C2
	MBAL3168
	ICS 1 WP04 HV Test Warm
	295.18

	12
	D1
	SSS003
	BEF3 CQW0 resistance measurement
	295.18

	10
	D2
	SSS107
	ICS 6 Disconnect Magnet WP07
	296.43

	4
	E1
	MBAL3141
	Prep 5 Pumping + Cool Down
	2.80 HV

	1
	E2
	MBBL2146
	PT 6.3 Training 3
	1.90

	5
	F1
	MBAR3285
	Prep 5 Pumping + Cool Down
	4.25 HV

	2
	F2
	MBBR3270
	PT 12 Warm Up
	53.7


Coordinator this week: J. Miles; MTM support on call :   16-3602 .

14  Magnets Tested   ( 1193 ???) 
10  dipoles( 2 with MM)+ 2  SSS(0  with MM) +  2  SSSS 
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
· Silver Shoes bench F etc
General Points 
e-Traveller & Paper Traveller

Problems   Power Converters  OK ?
Resistances & HV on SSS in 904

AOB

	


Tests 

	Magnets under Test from 06-02-2005 to 12-02-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA2
	MBAR3278 
	[image: image1.png]



	28-01-2005 
	1 h.
	23 h.
	16 h.
	22 h.
	13 h.
	07-02-2005 
	240 hours
	
	

	2
	TBA2
	MBAR3284 
	[image: image2.png]



	07-02-2005 
	33 h.
	39 h.
	38 h.
	15 h.
	9 h.
	12-02-2005 
	120 hours
	
	

	3
	TBB1
	SSS612 
	[image: image3.png]



	01-02-2005 
	44 h.
	27 h.
	37 h.
	16 h.
	20 h.
	07-02-2005 
	144 hours
	
	

	4
	TBB1
	SSS614 
	[image: image4.png]



	07-02-2005 
	18 h.
	24 h.
	11 h.
	12 h.
	7 h.
	10-02-2005 
	72 hours
	
	

	5
	TBB2
	MBBL1193 
	[image: image5.png]



	07-02-2005 
	-
	-
	-
	-
	-
	11-02-2005 
	96 hours
	
	

	6
	TBB2
	MBBL2141 
	[image: image6.png]



	02-02-2005 
	27 h.
	22 h.
	27 h.
	17 h.
	13 h.
	06-02-2005 
	96 hours
	
	

	7
	TBC1
	MBAR3281 
	[image: image7.png]



	03-02-2005 
	27 h.
	28 h.
	43 h.
	13 h.
	10 h.
	08-02-2005 
	120 hours
	
	

	8
	TBD1
	SSS127 
	[image: image8.png]



	05-02-2005 
	47 h.
	14 h.
	11 h.
	9 h.
	8 h.
	09-02-2005 
	96 hours
	
	

	9
	TBD2
	SSS098 
	[image: image9.png]



	03-02-2005 
	49 h.
	23 h.
	28 h.
	7 h.
	30 h.
	08-02-2005 
	120 hours
	
	

	10
	TBE1
	MBAR3279 
	[image: image10.png]



	08-02-2005 
	23 h.
	25 h.
	18 h.
	13 h.
	14 h.
	12-02-2005 
	96 hours
	
	

	11
	TBE1
	MBAR3282 
	[image: image11.png]



	03-02-2005 
	25 h.
	24 h.
	33 h.
	13 h.
	11 h.
	08-02-2005 
	120 hours
	
	

	12
	TBE2
	MBBR3215 
	[image: image12.png]



	06-02-2005 
	34 h.
	25 h.
	23 h.
	14 h.
	4 h.
	10-02-2005 
	96 hours
	
	

	13
	TBE2
	MBBR3245 
	[image: image13.png]



	02-02-2005 
	22 h.
	26 h.
	19 h.
	14 h.
	8 h.
	06-02-2005 
	96 hours
	
	

	14
	TBF1
	MBBR2525 
	[image: image14.png]



	05-02-2005 
	19 h.
	27 h.
	44 h.
	17 h.
	14 h.
	10-02-2005 
	120 hours
	
	

	15
	TBF2
	MBAL3104 
	[image: image15.png]



	05-02-2005 
	22 h.
	45 h.
	24 h.
	14 h.
	8 h.
	10-02-2005 
	120 hours
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR3278
	12048.1 A.
(D2-U S01)
	12296 A.
(D2_U_S01)
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR3284
	8459 A.
(D2_L)
	10884 A.
(D2_L)
	11726.5 A.
(D2-U)
	12310 A.
(D2_U)
	12367 A.
(D2-U)
	12665.3 A.
(D2-L)
	12765.2 A.
(D2-L)
	12634 A.
(D2_L)
	12807.8 A.
(D2_L)
	[image: image17.png]



	-
	-
	-

	3
	SSS612
	5482 A.
(E-P34)
	5820 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS614
	5820 A.
[image: image19.jpg]



	5820 A.
[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL1193
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	6
	MBBL2141
	11228 A.
(D1_U)
	12062 A.
(D2-L)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	7
	MBAR3281
	11436 A.
(D1_U)
	11721.9 A.
(D2-U)
	12179 A.
(D1 - U)
	12300 A.
(D1-U)
	12490.5 A.
(D1_U)
	12614 A.
(D1_L)
	12711 A.
(D1-U)
	12851 A.
(D2-L)
	12850 A.
[image: image23.jpg]



	-
	-
	-
	-
	

	8
	SSS127
	12063.6 A.
(QI_12-34)
	12228 A.
(QE_34)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	9
	SSS098
	11289 A.
(QI-34)
	0 A.
(No Data)
	12434.2 A.
(QE_34)
	[image: image25.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	10
	MBAR3279
	12149 A.
(D1-U)
	11657.3 A.
(D2-U)
	12738.8 A.
(D2-U)
	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	11
	MBAR3282
	9474 A.
(D1 - L)
	11697 A.
(D2_L)
	11925.7 A.
(D1_U)
	12246.1 A.
(D2_U)
	12600 A.
(D2-L)
	12798 A.
(D2-U)
	12837 A.
(D2 - U)
	12850 A.
[image: image27.jpg]



	-
	-
	-
	-
	-
	

	12
	MBBR3215
	11780 A.
(D2_U)
	12746 A.
(D1-L)
	[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	13
	MBBR3245
	10647 A.
(D1-U)
	11678 A.
(D2-L)
	12593 A.
(D1 - U)
	12850 A.
[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	14
	MBBR2525
	11618 A.
(D1-U-S01)
	12464 A.
(D2_L)
	[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	15
	MBAL3104
	12845 A.
(D1_L)
	12487 A.
(D1-L)
	12798.2 A.
(D1_L)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	


Week 7feb05 to 13Feb 05
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _08_M


	TBA2
	1.5KA provoked quench on 2123: For quench location shaft signals were not found.
	MTM
	
	Maryline called : Qloc cable not conned. Qantenna cable were in place. Qloc cables connected to the rack.

	Tuesday _08_M


	TBF1
	2525 magnet at 4 K. No HV test possible (interlock). Cannot go down to 1.9 K.

Quench at 4.2 K to warm up: SSL at 9170 A.
	ACR
	
	ACR team working on the CFB.PT142 pressure detector not working. Air leak found in safety valves which block the cool down. 10h lost. Magnet warm up without fast de-excitation test.

Wait for warming to repair completely. Reparation done Friday.

TC informed He advised to proceed with the warm up.



	Tuesday _08_A


	TBD1
	The Vtaps on E lead is broken, that was creating a trigger on the corresponding card of the auxiliairy POTAIM crate. 

	MTM
	
	 Short between double output 10 and 12 done by Maryline.

	Tuesday _08_A


	TBA1
	MM on 2123 magnet. Rotation became bad for the two apertures.


	ICS?
	
	TC informed. LHC cycle stopped. Coupling in the TRU broken. Repaired by Frederic Tuesday. Measurement repeated.

One solution to avoid such a problem: when rotating shafts manually push first the fast stop (which switches of the motors) of each aperture, and only then, rotate shafts carefully and slowly by hand to Qloc horizontal position.

	Tuesday _08_N

Thursday_10_M
	TBC1

TBF1
	During IAP warm_LF, interlocks: Mag300_x2 and Mag300air_x2 appear. TT821: 294.39K, cryo T is also 294K
Same problem.
	ACR
	
	Origin: cryo team do not put the bench in correct phase at the given time. Magnet has not to be  in the shutdown phase, the phase should be "connecting magnet".
Maryline asked ACR team to go in the good phase.

	Wednesday_09_M


	TBE2
	 SPA at 1000 A: quench at 850A
	AB/PO
	
	 Maryline called.Piquet Power Supply that solved the problem

	Wednesday_09_A


	TBD1
	TCQW0 (resistance measurement) lower voltage than the expected (at given current). Tsensor resistance value give also no the correct value.
	AB/CO
	IAP
	Resistance measurements and checks of cable give no striking results. 

	Wednesday_09_M


	TBA1
	Error in DB while generating HR Config"when Test was launched - LHC_Cycle_FT_Time 300s.

Shaft is not rotating correctly.
	AB/CO-MTM

MTM
	HWREC
	Maryline contacted. She advised to put off HWREC.



	Thursday _10_A


	TBE
	T_reading via video monitor not correct. 


	ACR
	
	Jerome corrected the readings

	Thursday _10_A


	TBB2
	Cryo leak on the 1193 magnet (leak in thermal screen)


	
	
	13h00 of stop. Magnet removed.               NC opened. Bottom tray to be changed.

	Thursday _10_A


	TBB1
	Scanning of the magnet name not possible.


	ICS
	
	Maryline found that the name of magnet on the label is not correct. Name entered manually.

	Friday _11_M

Friday _11_N


	TBC1
	HV on MB3286 .(MB/ground) failed : I leak exceed 20 micro A (30 micro A)

MB 3286: Quench at 4848 A. 
Test repeated: quench at 6680 A in the clamp
	ICS, ACR?
	
	TC leads the investigation: CFB insulation or bad connection.  TC asks to proceed with the tests carefully.

Investigation on the CFB will be done monday.

TC informed: he suggested increasing the He level to 80%.

Pierre asked to put the magnet in stand by till Saturday morning. TC asked to repeat the Quench: OK. Test went on normally.

Results :Insulation of the CFB correct. Insulation damaged at the end of Vtaps

	Friday _11_M


	TBC2
	Pumping to 1.9 K not possible. Purging of the system badly done.


	ACR
	
	6 h lost

	Friday _11_M


	TBD2
	Leak on the safety valve of the CFB


	ACR
	
	7 h lost

	Friday _11_M


	TBC2
	Server connectivity problem.

DAQ softwares not working


	IT?
	
	

	Friday _11_A


	TBC2
	Discharge of heaters when we the PC was switch on


	
	
	Solved by Maryline

	Friday _11_A


	TBC2
	During Diode test, heaters got discharged. No quench detected.
	
	
	Test repeated.

	Friday _11_A


	TBB1
	SSS034: IAP at warm initial: Errors in Temp sensor .
	
	
	T coefficient checked by Maryline : not correct. NC opened.

	Friday _11_A


	TBA2
	Training quench 9 on 3284: No quench detected by AQA. Signals seem OK.
	AB/CO
	AQA
	Boris fixed the problem.

	Friday _11_N


	TBB2
	IAP @ warm: Interlock Mag300_x2 and Mag300air_x2 appear.
	ACR
	
	Cryo team informed. They were performing tests and they will clear the interlock afterward. 

	Friday _11_N

Saturday_12_N


	TBD1

TBB1
	HV test on SSS032: condensation on Orbital dipoles.

HV test on SSS034  failed for the same reason
	MTM
	
	Fun connected to blow the air and evaporate the condensation.

Condensation on the terminals (fun used in another cluster?).

	Saturday_12_M


	TBD1
	QH test on SSS032: Alarm on tfire. 
	MTM
	
	Test repeated: same error. DAQ sampling faulty. Discharge recorded when the sampling frequency is changing. Proceed with the tests.

	Saturday_12_M


	TBA1
	MM: Shaft position alert at high field.
	MTM
	
	Maryline advised to continue: It can happen when the positioning is close to the acceptance at low field.

	Saturday_12_M


	TBD1
	Powering of CD circuit of SSS032  not possible: stand by error when switching the power supply.
	MTM
	
	Maryline checked Diagnostic: clamp Vtaps have probably been inversed during soldering.

	Saturday_12_M


	TBC2
	During the training of the 3253 analysis show that the diode is conducting. 
	MTM
	
	TC informed. He asked to check if the FPA is enable and advised to continue.



	Saturday_12_N


	TBC2
	De-excitation test cannot be launched.

Problem in arming lf/hf DQA during training
	MTM
	
	Maryline informed. She suggested resetting the crates, TEMA, the server. No success.

The wrong test was charged. Test performed manually.

	Sunday_13_M


	TBA
	MXI error : hf, lf , TEMA re-started
	AB/CO
	TEMA
	TEMA was re-started but correct parameters not re-loaded: D0502130039.ha01 and D05130550.ha2 must be transferred from MBAI123 to MBAI123_2.

	Sunday_13_M


	TBB1
	Powering of corrector not possible on SSS034
	AB/PO
	
	Maryline informed AB/PO on call. Problem with regulation loop.600 A reached in manual control

	Sunday_13_M


	TBD2
	Trigger on the CD circuit
	ICS?
	
	Maryline called  Diagnostic : Vtaps of the C clamp that is opened probably at the soldering place.

	Sunday_13_N


	TBE2
	Polarity Selector switch has problems at the end position while turning from "Right" to "Left" Polarity.
	MTM
	
	


 










